## **Author's Response To Reviewer Comments**

Clo<u>s</u>e

The manuscript has been attached as a .docx formatted file. PDF also attached.

## Editor Notes:

- 1) Please add the GigaDB DOI link to your reference [19] in the bibliography (http://dx.doi.org/10.5524/100645): Added to reference.
- 2) Please add a sentence at the end of your "software availability" section along the lines of: "An archival copy of the code, scripts and other supporting data are also available via the GigaScience database GigaDB [19]."

  Added in text.

Close